Author's response to reviews

Title: Human pluripotent embryonal carcinoma NTERA2 cl.D1 cells maintain their typical morphology in an angiomyogenic medium.

Authors:

Pedro D Simoes (pedroDanielsim@yahoo.co.uk)
Teresa Ramos (teresaramos@fmd.ul.pt)

Version: 3 Date: 28 March 2007

Author's response to reviews:

Dear sir,
We have noticed that a sentence referring to figure 2 was not correct, in page 5, 2nd paragraph, first line. Instead of "...in figure 2, after 5 days in culture..." it now reads "...in figure 2, after 20 days in culture..."
Best regards
Pedro Simoes and Teresa Ramos